



October 28, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 5636-review.doc).

Title: Post-translational modifications of hepatitis C viral proteins and their biological significance

Author: Jana Hundt, Zhubing Li, Qiang Liu

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 5636

The manuscript was reviewed by two reviewers. Both reviewers were very positive and did not raise any suggestions for revision.

As per the editorial comments, we have changed the format of the title page and references accordingly. We also added "Core tip" for our article.

A Copyright Assignment form signed by all authors is included as well.

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

A handwritten signature in black ink, appearing to be 'Qiang Liu'.

Qiang Liu, Ph.D.

Senior Scientist, Principal Investigator

Vaccine and Infectious Disease Organization-International Vaccine Center

University of Saskatchewan